Author's response to reviews

Title: Establishment and characterization of pleomorphic adenoma cell systems: an in-vitro demonstration of carcinomas arising secondarily from adenomas in the salivary gland.

Authors:

Satoshi Maruyama (maru@dent.niigata-u.ac.jp)
Jun Cheng (jun@dent.niigata-u.ac.jp)
Susumu Shingaki (shingaki@dent.niigata-u.ac.jp)
Takashi Tamura (ttamura@kyorin-u.ac.jp)
Shuichi Asakawa (asa@dmb.med.keio.ac.jp)
Shinsei Minoshima (mino@hama-med.ac.jp)
Yoshiko Shimizu (yshimizu@kyorin-u.asc.jp)
Nobuyoshi Shimizu (smzgsp@dmb.med.keio.ac.jp)
Takashi Saku (tsaku@dent.niigata-u.ac.jp)

Version: 8 Date: 19 July 2009

Author's response to reviews:

July 20, 2009

Dr. Melissa Norton, Editor-in-chief
Dr. The BioMed Central Editorial Production Team

MS: 1860293842287409

Dear Editors:

Thank you for your e-mail of July 13, 2009 with comments from the BioMed Central Editorial Production Team. We are grateful to them for their helpful remarks. Following the advice, we have checked and modified our manuscript. I am resubmitting our revised version of MS: 1860293842287409 entitled “Establishment and characterization of pleomorphic adenoma cell systems: an in-vitro demonstration of carcinomas arising secondarily from adenomas in the salivary gland.”

Yes, we have done this.

Sincerely yours,

Takashi Saku, DDS, PhD
Division of Oral Pathology
Department of Tissue Regeneration and Reconstruction